Author’s response to reviews

Title: Realities of replication: implementation of evidence-based interventions for HIV prevention in real-world settings

Authors:

Shayna D Cunningham (shayna.cunningham@gmail.com)
Josefina J Card (jjcard@socio.com)

Version: 3 Date: 30 December 2013

Author’s response to reviews:

Dear Implementation Science Editorial Team,

Attached is a clean version of our manuscript titled Realities of replication: implementation of evidence-based interventions for HIV prevention in real-world settings (MS: 4407680231013247). We accepted all of the changes suggested by the copy-editor. Please let us know if you have any additional questions regarding this manuscript.

Kind regards,

Shayna Cunningham